Author's response to reviews

Title: Managing depression in primary care: A meta-synthesis of qualitative and quantitative research from the UK to identify barriers and facilitators.

Authors:

Elizabeth A Barley (elizabeth.barley@iop.kcl.ac.uk)
Joanna Murray (joanna.murray@iop.kcl.ac.uk)
Paul Walters (paul.walters@iop.kcl.ac.uk)
Andre Tylee (andre.tylee@iop.kcl.ac.uk)

Version: 5 Date: 6 June 2011

Author's response to reviews:

6th June 2011
Thank you for your email. Please accept my apologies for a late reply, as per my previous correspondence, I have been on annual leave and unable to access email. We have dealt with your requested changes as follows:

Major revisions (we require the author to make these changes)

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution - completed.

References: Please make sure references are cited in numerical order. We notice reference 65 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited. – as noted in our previous correspondence, this reference is not in the text at all but cited in Table 1 (reference to an outcome measure). We have therefore cited this at the end of the references according to your ‘author instructions’. Please let us know if this is not correct and if so, how you would like us to address this.

Appendix: please format this as normal text (not in a table) and move both parts into the main body of the text after the 'Conclusions'. - completed

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Abstract: please remove the manuscript title from above the abstract. - completed

White space: please reduce the spacing between sections to minimise white space. e.g. page 24 is a blank page. – we have removed any ‘spare’ white space and lines.
Authors’ contributions: please ensure this is formatted into a paragraph rather than a list or in bullet points. - completed

Coloured text: please remove all of this. - completed

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures. - we have cropped the figure as much as possible without distorting it.

We trust this is satisfactory, although please advise us if we need to do something about the references (point 2 above). We look forward to receiving your response.

Kind regards

Elizabeth Barley